Author’s response to reviews

Title: Predictors of orthorexic behaviours in patients with eating disorders. A preliminary study

Authors:

Anna Brytek-Matera (abrytek-matera@swps.edu.pl)
Radoslaw Rogoza (rogoza.radoslaw@gmail.com)
Carla Gramaglia (carla.gramaglia@gmail.com)
Patrizia Zeppegno (patrizia.zeppegno@med.uniupo.it)

Version: 2 Date: 28 Sep 2015

Author’s response to reviews:

Dear Professor Fassino,

We would like to thank you for your decision.

The changes to the manuscript are indicated in the text by highlighting (in green).

The manuscript id: BPSY-D-15-00089

Please be informed that the editor is happy to your revision as he stated above. And please provide the following editorial requests prior to acceptance below.

1. Please provide full name of Ethics Committee.

   We added into the text information about full name of Ethics Committee (the SWPS University of Social Sciences and Humanities Human Research Ethics Committee) (page 5)

2. Please specifically state the informed consent was obtained, not just asked for.

   We added information about oral and written informed consent into the text (page 5).